Author’s response to reviews

Title: Case Report: shock septic from multidrug resistant descending necrotizing mediastinitis - intensive multimodal approach IgM enriched immunoglobulins and Polymyxin B hemoperfusion

Authors:
Vincenzo Pota (vincenzo.pota@unicampania.it)
Maria Beatrice Passavanti (beatrice.passavanti@libero.it)
Pasquale Sansone (pasquale.sansone@unicampania.it)
Maria Caterina Pace (caterina.pace@libero.it)
Filomena Peluso (filomena.peluso@gmail.com)
Alfonso Fiorelli (alfonso.fiorelli@unicampania.it)
Caterina Aurilio (caterina.aurilio@unicampania.it)

Version: 2 Date: 07 Nov 2017

Author’s response to reviews:

Title: Case Report: Shock Septic from descending necrotizing mediastinitis - combined treatment with IgM enriched immunoglobulin preparation and direct polymixin B hemoperfusion

Journal: Journal of Medical Case Reports

Dear Reviewer,

It’s a pleasure for me to practice all the correction required

Reviewer 2

4c: I’ve cleared the timing of the patient history and I’ve added a timeline figure
5. I’ve shortened the discussion and eliminated the re-writing

8. I’ve sent the manuscript to a native English writer who corrected all.

9. I’ve explained the IgM enrichment rationale. I’ve reorganized the table

Finally I’ve corrected the capital letters and added the normal range to the parameters to the text. I’ve also cleared the 2.8 and 3.8 tooth avulsion and the …false pneumonia…

I’ve explained MOF and SOFA

Reviewer 3

2. I’ve corrected ethical concerns

3. I’ve modified the introduction as you suggested

4. I’ve modified as you suggested.

c. I’ve cleared the timing of the patient history and I’ve added a timeline figure

e. I’ve cleared the mechanism of intervention

I’ve reorganized the discussion as you suggested.

I’ve corrected table 1
I’ve removed fig 3,4
I’ve corrected the abstract
I’ve checked and corrected all grammatical and spelling errors.

I wish it will be all ok

Sincerely

Vincenzo Pota